Author's response to reviews

Title: Should socioeconomic indicators be included in burden of disease estimates? An example illustrating the burden of type 2 diabetes attributed to lower educational levels in Sweden

Authors:

Emilie E Agardh (emilie.agardh@ki.se)
Anna Sidorchuk (anna.sidorchuk@ki.se)
Johan Hallqvist (johan.hallqvist@ki.se)
Rickard Ljung (rickard.ljung@ki.se)
Stefan Peterson (stefan.peterson@ki.se)
Tahereh Moradi (tahereh.moradi@ki.se)
Peter Allebeck (peter.allebeck@ki.se)

Version: 4 Date: 31 October 2011

Author's response to reviews: see over
Dear Dr. Murray and Dr. Lopez,

We hereby re-submit the enclosed manuscript “Should socioeconomic indicators be included in burden of disease estimates? An example illustrating the burden of type 2 diabetes attributed to lower educational levels in Sweden”, with a view for publication in the Population Health Metrics.

We have answered the editors comments one by one and made changes accordingly (see below).

Competing interest: none declared.

Sincerely yours,

Emilie Agardh, RN, MPH, PhD
On behalf of all co-authors

Emilie Agardh, RN, MPH, PhD
Address: Karolinska Institutet, Department of Public Health Sciences, Division of Social Medicine, Norrbacka Plan 6, Karolinska Hospital
SE-171 76 Stockholm, Sweden
E-mail: Emilie.agardh@ki.se
Phone number: +46-765519941
Fax number: +46-8-300333
We thank the editors for the valuable comments that we have answered one by one.

<table>
<thead>
<tr>
<th>Comments from editors</th>
</tr>
</thead>
<tbody>
<tr>
<td>Editors</td>
</tr>
<tr>
<td>Answer from authors</td>
</tr>
<tr>
<td>Editors</td>
</tr>
<tr>
<td>Answer from authors</td>
</tr>
<tr>
<td>Editors</td>
</tr>
<tr>
<td>Answer from authors</td>
</tr>
</tbody>
</table>